Author's response to reviews

Title: Primary PEComa of the bladder treated with primary excision and adjuvant interferon-alpha immunotherapy: a case report

Authors:

Jeremy R Parfitt (jrparfit@uwo.ca)
Anthony J Bella (ABella@urology.ucsf.edu)
Bret M Wehrli (bret.wehrli@lhsc.on.ca)
Jonathan I Izawa (jonathan.izawa@lhsc.on.ca)

Version: 4 Date: 18 August 2006

Author's response to reviews:

1. Title and headings changed to sentence case.
2. Affiliations written out in full, according to requested format.
3. "PAS" spelled out as Periodic acid-Schiff (page 4).
4. "GI" spelled out as gastrointestinal upon first time use (page 5).
5. "This group of tumors" changed to "These tumors" (page 5).
6. "interferon" changed to "IFN" (page 7).
7. Rare addition/ommission of punctuation marks.